Author's response to reviews

Title: Peripheral Arterial Disease Study (PERART): Prevalence and predictive values of asymptomatic peripheral arterial occlusive disease related to cardiovascular morbidity and mortality.

Authors:

M Teresa Alzamora (mtalzamora.bnm.ics@gencat.net)
Jose Miguel Baena-Diez (jbaenad@meditex.es)
Marta Sorribes (martasorribes.pbcn@ics.scs.es)
Rosa Fores (rfores.bnm.ics@gencat.net)
Pere Toran (ptoran.bnm.ics@gencat.net)
Marisa Vicheto (mvicheto@gmail.com)
Guillem Pera (gpera.bnm@ics.scs.es)
M Dolores Reina (dreina.bnm.ics@gencat.net)
Carles Albaladejo (24270cab@comb.es)
Judith Llussa (jllussa.bnm.ics@gencat.net)
Magda Bundo (mbundo.bnm.ics@gencat.net)
Amparo Sancho (asanchop.bnm.ics@gencat.net)
Joan Rubies-Prat (jrubiesp@searteriosclerosis.org)
Juan Francisco Arenillas (juanfarenillas@gmail.com)
Antonio Heras (aheras.bnm.ics@gencat.net)

Version: 3 Date: 29 November 2007

Author's response to reviews: see over
Dear Sir/Madame;

Please find attached the revised manuscript (4262758615513712) entitled “Peripheral Arterial Disease Study (PERART): Prevalence and predictive values of asymptomatic peripheral arterial occlusive disease related to cardiovascular morbidity and mortality”.

According to your suggestion we have made the following changes:

We have removed “BMC Public Health” and “Study Protocol” from the title page.

In the title, we have omitted the full stop at the end of the title.

We have modified each affiliation as you suggested: department, institute, city and country.

We have changed "Methods to Methods/Design”.

We have declared: “The authors declare that they have no competing interests”.

In the references, we have listed all the authors. We have removed “et al”.

We have checked that all references are cited in order, in square brackets and before the full stop.

Thank you very much for your kind attention. I am looking forward to hearing from you.

Yours sincerely,

MT Alzamora Sas
Unitat de Suport a la Recerca
Àmbit Barcelonés Nord i Maresme
CAP El Maresme
C/ Camí del Mig, 36
08303-Mataró
Barcelona (Spain)

Phone: +34 93 741 53 38
Fax: +34 93 799 93 17
E-mail: maiteal2007@gmail.com